Open Access Full Text Article

SHORT REPORT

# Use of Clinical Global Impressions-Severity (CGI-S) to Assess Response to Antidepressant Treatment in Patients with Treatment-Resistant Depression

Joachim Morrens<sup>1</sup>, Maju Mathews<sup>2</sup>, Vanina Popova<sup>1</sup>, Stephane Borentain 60<sup>2</sup>, Benoit Rive<sup>3</sup>, Beatriz Gonzalez Martin Moro<sup>4</sup>, Carol Jamieson<sup>5</sup>, Qiaoyi Zhang<sup>2</sup>

<sup>1</sup>Janssen Research & Development, Beerse, Belgium; <sup>2</sup>Janssen Global Services, LLC, Titusville, NJ, USA; <sup>3</sup>Janssen-Cilag, Paris, France; <sup>4</sup>Janssen-Cilag, Madrid, Spain; <sup>5</sup>Janssen Research & Development, LLC, Milpitas, CA, USA

Correspondence: Qiaoyi Zhang, Janssen Global Services, LLC, 1125 Trenton-Harbourton Road, Titusville, NJ, 08560, USA, Tel +1 908 300 2500, Email qzhang87@its.jnj.com

Background: This post-hoc analysis evaluated the agreement between Clinical Global Impressions-Severity (CGI-S) score- and Montgomery-Åsberg Depression Rating Scale (MADRS) total score-based assessment of response in patients with treatment-resistant depression (TRD) treated with esketamine nasal spray plus a newly initiated oral antidepressant (ESK-NS + AD).

Methods: Data were analyzed from a phase 3, randomized, double-blind study (TRANSFORM-2) of flexibly dosed esketamine or placebo nasal spray plus a newly initiated oral-AD in adults with moderate-to-severe TRD. Patients with ≥50% reduction in MADRS from baseline at the end of the 4-week acute treatment phase were defined as responders. For the CGI-S-based assessment of response, patients with  $\geq 2$  points decrease from baseline or a CGI-S score of  $\leq 3$  (mildly depressed to normal) were considered responders. Cohen's kappa coefficient was calculated to assess level of agreement between MADRS and CGI-S-based assessments.

**Results:** At the end of 4-week treatment, the proportion of responders among all study patients (n=201) was similar when assessed using the MADRS (61%) and CGI-S (62%) methods, with substantial agreement (Cohen's kappa=0.76; sensitivity=92%; specificity=84%) between both methods. When restricting analysis to ESK-NS + AD-treated patients (n=101) who had a higher response rate (on MADRS: 69%; on CGI-S: 68%), the agreement remained substantial (Cohen's kappa=0.75; sensitivity=91%; specificity=84%).

Conclusion: The CGI-S may be a practical and reliable alternative to the MADRS to assess response to ESK-NS + AD in patients with TRD and can be used in real-world practice to support informed treatment decisions.

Keywords: antidepressant response, Clinical Global Impressions-Severity, esketamine, Montgomery-Asberg Depression Rating Scale, treatment-resistant depression

#### Introduction

Measurement-based care (MBC) is a critical component of an evidence-based decision support framework that guides patient treatment. In mental health clinical practice, MBC approaches have been designed to provide systematic, quantitative assessments to track treatment progress, measure symptom severity and reflect on important aspects of health-related quality of life, satisfaction and functioning.<sup>2,3</sup> Thus, MBC can be implemented in various ways to enhance clinical practice in mental healthcare settings. Several clinician-rated instruments that rely on interviews and questionnaires for assessments of symptoms have been used to measure treatment outcomes in clinical studies of patients with major depressive disorder (MDD).<sup>4,5</sup> The Montgomery-Åsberg Depression Rating Scale (MADRS), a 10-item questionnaire (each item yields a score of 0 to 6), is a validated, clinician-rated measure of depression designed to measure depression severity and to detect changes due to antidepressant treatment in patients with MDD.<sup>6</sup> It is one of the most frequently used scales for assessment of depression in clinical trials of antidepressant treatments submitted to the US Food and Drug Administration (FDA) and the European Medicines Agency (EMA).<sup>5,7</sup> However, the MADRS as

Morrens et al Dovepress

a research instrument usually takes approximately 20–40 minutes to administer and therefore may not be feasible for clinicians to use in their routine practice.<sup>6,8</sup>

By contrast, the Clinical Global Impressions-Severity (CGI-S) is a single-item, 7-point scale designed to assess global impression of severity rather than rating individual depressive symptoms. The CGI-S requires the clinician to rate the severity of patient's illness from a global perspective, relative to the clinician's experience with patients who have MDD. The brevity and simplicity of CGI-S makes it a more practical measurement tool that can easily be used by a clinician in a busy real-world clinical practice setting. The CGI-S directly reflects a clinician's judgment and captures clinical impressions of depression severity based on observed and reported symptoms, behavior, and function in the past 7 days. 9,10 The scale has also shown good inter-rater reliability among clinicians. 7,9,11 Several studies have evaluated the association between change on the MADRS and a 1-point change on the CGI-S and report a range of results. A 1-point reduction in CGI-S score translated into an 8-9-point change in MADRS total score, in a study evaluating the relationship of MADRS and CGI-S scales in patients with MDD. Another study concluded that based on a 1-point improvement in the CGI-S anchor, a 10-point improvement on the MADRS is an appropriate meaningful change threshold for withinpatient change. 12 In a recent post-hoc analysis of phase 3 studies of esketamine nasal spray in patients with treatmentresistant depression (TRD), a 2-point reduction in CGI-S score corresponded to a 12-point reduction in MADRS, thus establishing thresholds for clinically meaningful within-patient changes and better interpretability. 13 Phase 3 clinical studies of esketamine nasal spray in patients with TRD have used MADRS as the primary assessment of treatment response and remission. 14-18 Both MADRS and CGI-S scales have demonstrated comparable ability to measure treatment outcomes with a high correlation in patients with MDD.<sup>4</sup>

Evaluating rating agreement between MADRS-based and CGI-S-based outcomes using patient-level data is important to translate clinical trial results into clinical practice. The purpose of this post-hoc analysis was to evaluate the agreement with the pre-specified MADRS-based assessment and a CGI-S score-based definition of response to acute antidepressant treatment in patients with TRD.

#### **Methods**

# Study Design and Patients

Data for this post-hoc analysis were derived from a phase 3, randomized, double-blind, active-controlled study (TRANSFORM-2; NCT02418585)<sup>17</sup> of flexibly dosed esketamine nasal spray in patients with TRD. Details of study design and population have been described elsewhere.<sup>17</sup> Briefly, patients with MDD aged between 18 and 64 years with moderate-to-severe depression (30-item Inventory of Depressive Symptoms-Clinician-rated total score  $\geq$ 34) were included. Patients also had to meet the study definition of TRD: failed to respond to  $\geq$ 2 antidepressant treatments of adequate dose and duration in the current episode, with one antidepressant treatment failure assessed prospectively at study entry. The study had three phases: a 4-week screening/prospective observation phase, a 4-week double-blind treatment phase, and a posttreatment follow-up phase of up to 24 weeks. In the 4-week acute treatment phase, patients were randomly assigned (1:1) to treatment with flexibly dosed esketamine nasal spray (56 or 84 mg) twice a week plus a newly initiated daily oral antidepressant (ESK-NS + AD) or oral antidepressant plus placebo nasal spray (AD + PBO-NS).

Independent Ethics Committees or Institutional Review Boards (listed in <u>Supplementary Material</u>) for each country approved the study protocol. The study was conducted in accordance with ethical principles of the Declaration of Helsinki, Good Clinical Practices and applicable regulatory requirements. All patients provided written informed consent before study enrollment.

#### Assessments

The primary efficacy endpoint of the TRANSFORM-2 study was change in MADRS total score from baseline (day-1) to endpoint (day-28) of the double-blind treatment phase. Since transient dissociative and sedative effects of esketamine nasal spray could potentially bias the rater, all MADRS assessments were performed predose by independent, remote (by telephone) raters who were blind to the protocol details. Response (secondary endpoint) was pre-specified as patients

Dovepress Morrens et al

with  $\geq$ 50% reduction from baseline in MADRS total scores. Investigators at the site rated the severity of depression using the CGI-S. CGI-S assessments were also conducted predose.

For the alternative CGI-S-based assessment of response in this post-hoc analysis, patients with  $\geq 2$  points decrease from baseline or a CGI-S score of  $\leq 3$  (mildly depressed to normal) were considered responders based on the cut-off described previously in the literature. As per CGI-S guidelines, based on severity of illness of the patient, score of 1 to 2 was assigned, wherein, 1 = Normal – not at all ill; 2 = Borderline ill; 3 = Mildly ill; 4 = Moderately ill; 5 = Markedly ill; 6 = Severely ill; and 7 = Among the most extremely ill patients. The proposed cut-off criterion was based on a best fit model of optimal definitions of response from the literature.

#### Statistical methods

Cohen's kappa coefficient was calculated to assess the level of agreement between MADRS- and CGI-S-based assessments of response. Sensitivity and specificity of the CGI-S-determined response were estimated, using MADRS-based classification as reference. As part of additional sensitivity analyses, the level of agreement between MADRS- and other definitions of CGI-S-based response (2-point improvement or a CGI-S score of  $\leq$ 3) was also performed.

#### Results

#### **Patients**

Details of patient demographics, baseline characteristics and disposition have been provided in the primary study publication.<sup>17</sup> The dataset for the efficacy analyses of the TRANSFORM-2 study included 223 patients (ESK-NS + AD: n=114; AD + PBO-NS: n=109) who had a mean (SD) age of 45.7 (11.89) years and were mostly women (61.9%) and white (93.3%). At baseline, the mean (SD) MADRS total score was 37.1 (5.67), CGI-S total score was 5.1 (0.67) and all patients with moderate-to-severe TRD had a CGI-S score ranging from 4 to 7.

## Assessment of Response

A total of 201 patients had a day-28 MADRS assessment; however, 5 patients did not have a day-28 CGI-S assessment. Also, of the 198 patients who had a day-28 CGI-S assessment, two did not have a day-28 MADRS assessment. Thus, a total of 196 patients with both MADRS and CGI-S at day-28 were included in the analysis. At the end of the 4-week double-blind phase, the proportion of responders among all patients (n=201) was similar when assessed using the MADRS (61%) and CGI-S (63%) methods (Table 1).

The Cohen's kappa was 0.76 (95% CI: 0.67–0.86), suggesting substantial agreement between the two methods. The CGI-S-based assessment demonstrated 92.5% sensitivity and 82.9% specificity versus the MADRS-based assessment. These findings were further supported by additional sensitivity analysis, wherein CGI-S-based response using both definitions showed comparable agreement with the MADRS-based response assessment. In the overall population, the composite definition showed better sensitivity and slightly higher agreement with MADRS than the one based on 2-point

	All Patients <sup>a</sup> (N=196)		Esketamine Nasal Spray Plus Oral Antidepressant (N=100)	
	MADRS	CGI-S <sup>c</sup>	MADRS <sup>b</sup>	CGI-S <sup>c</sup>
Responders, n (%)	120 (61.2)	124 (63.3)	69 (69.0)	68 (68.0)
Non-responders, n (%)	76 (38.8)	72 (36.7)	31 (31.0)	32 (32.0)
Sensitivity, %	92.5		91.3	
Specificity, %	82.9		83.9	
Cohen's kappa (95% CI)	0.76 (0.67–0.86), p<0.0001		0.75 (0.60–0.89), p<0.0001	

Table I Assessment of Response Using MADRS and CGI-S

Notes: <sup>a</sup>Includes patients who received esketamine nasal spray plus oral antidepressant and patients who received oral antidepressant plus placebo nasal spray; <sup>b</sup>Patients with ≥50% reduction in MADRS from baseline at the end of the 4-week acute treatment phase were defined as responders; <sup>c</sup>Patients with ≥2 points decrease from baseline or a CGI-S score of ≤3 (mildly depressed to normal) were defined responders.

Abbreviations: CGI-S, Clinical Global Impressions-Severity; CI, confidence interval; MADRS, Montgomery-Asberg Depression Rating Scale.

Morrens et al **Dove**press

improvement (Cohen's kappa [95% CI] = 0.73 [0.63–0.83]), while the one based on CGI-S <3 showed marginally higher Cohen's kappa (0.80 [0.71–0.88]) (Supplementary Table 1).

Similar results were observed in patients treated with ESK-NS + AD (Table 1). The proportion of responders was 69% using the MADRS scale and 68% using the CGI-S scale. There was substantial agreement (Cohen's kappa=0.75, 95% CI: 0.60–0.89) between the two scales and the CGI-S assessment showed 91.3% sensitivity and 83.9% specificity. Additional analyses of CGI-S-based response using the other two definitions also showed comparable agreement with MADRS (Supplementary Table 1).

#### Discussion

This post-hoc analysis confirmed that the pre-specified MADRS criterion for response, ie, ≥50% reduction from baseline in MADRS total scores, used to evaluate acute treatment effect of ESK-NS + AD substantially corresponded with the CGI-S definition of  $\geq 2$  points reduction from baseline or cut-off score of  $\leq 3$ . At the end of the 4-week treatment phase, the proportions of responders and non-responders among all patients and patients treated with ESK-NS + AD assessed using the MADRS response criterion were similar to those assessed using the CGI-S criterion. There was substantial agreement (Cohen's kappa coefficient: 0.75-0.76) between the two scales.

Standardized scales provide tangible evidence of treatment response in clinical research. However, the implementation of scales such as MADRS in real-world psychiatry practice is often burdensome for the clinicians and patients. Given the variety of rating scales, using clinically and practically relevant measures that could potentially lower the burden of time associated with assessment is of great importance. 19,20 The MADRS has been used in pivotal clinical studies of antidepressants; however, its adoption in clinical practice is uncommon. The purpose of this study was to evaluate an alternative approach to the MADRS using the CGI-S that has similar rigor for assessment of response albeit with lesser burden and can be used by clinicians in real-world practice. During routine clinical visits, the use of a semistructured MADRS tool may not be practical to the physician and cumbersome for patients who have to formally answer to 10 standardized questions and can potentially restrict patient-physician interaction to an assessment bound within the 10 items of the scale. Findings from the present analysis underscore the agreement between depression rating scales and measures of global illness severity and indicate a high agreement between the MADRS and CGI-S for assessment of response. Thus, the simpler and readily understood CGI-S mimics the assessment of response in the clinical trial and can be used to measure treatment response in real-world clinical practice. 9 By utilizing a clinician-determined summary, the CGI-S is more intuitive and improvement on the CGI-S reflects a more global account of the patient's disease status and overall well-being, which is clinically meaningful for the clinician, patients and payer in contrast to assessments based on standardized rating scales for treatment effectiveness.9

This post-hoc analysis was based on data from over 200 adult patients with the same diagnosis (and definition) of TRD participating in a phase 3 clinical study from the registration program of esketamine nasal spray, representing a relatively homogenous population for assessment.<sup>17</sup> Furthermore, the analysis included data from both active control (AD + PBO-NS)- and study drug (ESK-NS + AD)-treated patients, providing wider range scores that make the findings conceivably more relevant and robust.

#### Limitations

Inherent limitations of a post-hoc analysis should be considered. The primary study enrolled patients with limited medical and psychiatric co-morbidities and substance use disorder history. Patients with current or recent history (within 6 months) of homicidal ideation/intent or suicidal ideation with intent to act or suicidal behavior (within the past year); diagnosis of psychotic disorder, MDD with psychotic features, bipolar or related disorders, borderline, antisocial, histrionic, or narcissistic personality disorder, obsessive-compulsive disorder, intellectual disability, autism spectrum disorder; uncontrolled hypertension; seizures were excluded; thus, the sample may not be representative of all the patients with TRD. However, a majority of the study population had either moderate or serious anxiety at baseline, similar to the broader population with TRD. The use of CGI-S also has some inherent limitations. Standardization across clinics typically does not occur; so, neither interrater reliability nor a consensus of how much each rating relies on symptoms or daily function or satisfaction has been agreed. Furthermore, given the

Dovepress Morrens et al

concerns of clinician bias, a brief patient-reported scale would be a suitable alternative for regular in-clinic assessments. Nevertheless, the CGI-S approach reflects in many ways the current assessment process in real-world practice and in this case allows the definition of clinically meaningful changes. Finally, it is acknowledged that the most widely used approach of considering a reduction of severity of symptoms of ≥50% from baseline to be a clinical responder is to a large extent arbitrary.<sup>13</sup> Moreover, a report based on data from patients with TRD found that a 35% reduction in the MADRS score was associated with a clinically meaningful degree of improvement in health-related quality-of-life<sup>21</sup> as assessed by the Quality-of-Life Enjoyment and Satisfaction Questionnaire.<sup>22</sup>

#### **Conclusions**

To the best of our knowledge, this analysis is the first to report comparable assessment of response using MADRS- and CGI-S-based criteria following acute treatment with esketamine nasal spray plus an oral antidepressant or oral antidepressant plus placebo nasal spray in patients with TRD. CGI-S is a practical and reliable alternative for MADRS to assess treatment response and therefore the clinicians can use the CGI-S-based definition to evaluate response to an antidepressant treatment, including esketamine nasal spray in patients with TRD in real-world practice to support informed treatment decisions.

## **Data Sharing Statement**

The data sharing policy of Janssen Pharmaceutical Companies of Johnson & Johnson is available at <a href="https://www.janssen.com/clinical-trials/transparency">https://www.janssen.com/clinical-trials/transparency</a>. Requests for access to the study data can be submitted through Yale Open Data Access (YODA) Project site at <a href="http://yoda.yale.edu">http://yoda.yale.edu</a>.

## **Acknowledgments**

Priya Ganpathy, MPharm CMPP (SIRO Clinpharm Pvt. Ltd., India) provided writing assistance and Harry Ma, PhD CMPP (Janssen Global Services, LLC) provided additional editorial support. The authors also thank the patients and investigators for their participation in the study. Previously presented as an e-poster at the ISPOR Annual Meeting 2020 (Virtual), May 18–20, 2020.

#### **Author Contributions**

All authors made a significant contribution to the work reported, whether that is in the conception, study design, execution, acquisition of data, analysis and interpretation, or in all these areas; took part in drafting, revising or critically reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work.

# Funding

This study was supported by Janssen Research & Development, LLC.

#### **Disclosure**

All authors are employees of Janssen and may hold company stocks or stock options. Joachim Morrens was an employee of Johnson & Johnson during the conduct of the study. The authors report no other conflicts of interest in this work.

#### References

- 1. Scott K, Lewis CC. Using measurement-based care to enhance any treatment. Cogn Behav Pract. 2015;22(1):49–59. doi:10.1016/j. cbpra.2014.01.010
- Trivedi MH, Daly EJ. Measurement-based care for refractory depression: a clinical decision support model for clinical research and practice. *Drug Alcohol Depend*. 2007;88(Suppl 2):S61–S71. doi:10.1016/j.drugalcdep.2007.01.007
- 3. Hong RH, Murphy JK, Michalak EE, et al. Implementing measurement-based care for depression: practical solutions for psychiatrists and primary care physicians. *Neuropsychiatr Dis Treat*. 2021;17:79–90. doi:10.2147/NDT.S283731
- Jiang Q, Ahmed S. An analysis of correlations among four outcome scales employed in clinical trials of patients with major depressive disorder. Ann Gen Psychiatry. 2009;8(1):4. doi:10.1186/1744-859X-8-4

Morrens et al **Dove**press

5. Khan A, Khan SR, Shankles EB, Polissar NL. Relative sensitivity of the Montgomery-Asberg Depression Rating Scale, the Hamilton Depression Rating Scale and the Clinical Global Impressions rating scale in antidepressant clinical trials. Int Clin Psychopharmacol. 2002;17(6):281–285. doi:10.1097/00004850-200211000-00003

- 6. Montgomery SA, Asberg M. A new depression scale designed to be sensitive to change. Br J Psychiatry. 1979;134(4):382-389. doi:10.1192/
- 7. Leucht S, Fennema H, Engel RR, Kaspers-Janssen M, Lepping P, Szegedi A. What does the MADRS mean? Equipercentile linking with the CGI using a company database of mirtazapine studies. J Affect Disord. 2017;210:287–293. doi:10.1016/j.jad.2016.12.041
- 8. Kearns NP, Cruickshank CA, McGuigan KJ, Riley SA, Shaw SP, Snaith RP. A comparison of depression rating scales. Br J Psychiatry. 1982;141 (1):45-49. doi:10.1192/bjp.141.1.45
- 9. Busner J, Targum SD. The clinical global impressions scale: applying a research tool in clinical practice. Psychiatry. 2007;4(7):28-37.
- 10. Quilty LC, Robinson JJ, Rolland JP, Fruyt FD, Rouillon F, Bagby RM. The structure of the Montgomery-Asberg depression rating scale over the course of treatment for depression. Int J Methods Psychiatr Res. 2013;22(3):175-184. doi:10.1002/mpr.1388
- 11. Jones RM, Patel K, Moscovici M, McMaster R, Glancy G, Simpson AIF. Adaptation of the clinical global impression for use in correctional settings: the CGI-C. Front Psychiatry. 2019;10:687. doi:10.3389/fpsyt.2019.00687
- 12. Hudgens S, Floden L, Blackowicz M, et al. Meaningful change in depression symptoms assessed with the Patient Health Questionnaire (PHQ-9) and Montgomery-Asberg Depression Rating Scale (MADRS) among patients with treatment resistant depression in two, randomized, double-blind, active-controlled trials of esketamine nasal spray combined with a new oral antidepressant. J Affect Disord. 2021;281:767–775. doi:10.1016/j. iad.2020.11.066
- 13. Turkoz I, Alphs L, Singh J, et al. Clinically meaningful changes on depressive symptom measures and patient-reported outcomes in patients with treatment-resistant depression. Acta Psychiatr Scand. 2021;143(3):253-263. doi:10.1111/acps.13260
- 14. Daly EJ, Trivedi MH, Janik A, et al. Efficacy of esketamine nasal spray plus oral antidepressant treatment for relapse prevention in patients with treatment-resistant depression: a randomized clinical trial. JAMA Psychiatry. 2019;76(9):893. doi:10.1001/jamapsychiatry.2019.1189
- 15. Fedgchin M, Trivedi M, Daly EJ, et al. Efficacy and safety of fixed-dose esketamine nasal spray combined with a new oral antidepressant in treatment-resistant depression: results of a Randomized, Double-Blind, Active-Controlled Study (TRANSFORM-1). Int J Neuropsychopharmacol. 2019;22(10):616-630. doi:10.1093/ijnp/pyz039
- 16. Ochs-Ross R, Daly EJ, Zhang Y, et al. Efficacy and safety of esketamine nasal spray plus an oral antidepressant in elderly patients with treatment-resistant depression-TRANSFORM-3. Am J Geriatr Psychiatry. 2019;28(2):121-141. doi:10.1016/j.jagp.2019.10.008
- 17. Popova V, Dalv EJ, Trivedi M, et al. Efficacy and safety of flexibly dosed esketamine nasal spray combined with a newly initiated oral antidepressant in treatment-resistant depression: a Randomized Double-Blind Active-Controlled Study. Am J Psychiatry. 2019;176(6):428-438. doi:10.1176/appi.ajp.2019.19020172
- 18. Wajs E, Aluisio L, Holder R, et al. Esketamine nasal spray plus oral antidepressant in patients with treatment-resistant depression: assessment of long-term safety in a Phase 3, Open-Label Study (SUSTAIN-2). J Clin Psychiatry. 2020;81(3). doi:10.4088/JCP.19m12891
- 19. Harding KJ, Rush AJ, Arbuckle M, Trivedi MH, Pincus HA. Measurement-based care in psychiatric practice: a policy framework for implementation. J Clin Psychiatry. 2011;72(8):1136-1143. doi:10.4088/JCP.10r06282whi
- 20. Waldrop J, McGuinness TM. Measurement-based care in psychiatry. J Psychosoc Nurs Ment Health Serv. 2017;55(11):30–35. doi:10.3928/ 02793695-20170818-01
- 21. Conway CR, Kumar A, Xiong W, Bunker M, Aaronson ST, Rush AJ. Chronic vagus nerve stimulation significantly improves quality of life in treatment-resistant major depression. J Clin Psychiatry. 2018;79(5). doi:10.4088/JCP.18m12178
- 22. Gaynes BN, Asher G, Gartlehner G, et al. Definition of treatment-resistant depression in the medicare population. Rockville (MD); 2018.
- 23. Guy W. ECDEU Assessment Manual for Psychopharmacology. Rockville, MD, USA: US Department of Health, Education, and Welfare; 1976:217-222.

#### Neuropsychiatric Disease and Treatment

# Dovepress

## Publish your work in this journal

Neuropsychiatric Disease and Treatment is an international, peer-reviewed journal of clinical therapeutics and pharmacology focusing on concise rapid reporting of clinical or pre-clinical studies on a range of neuropsychiatric and neurological disorders. This journal is indexed on PubMed Central, the 'PsycINFO' database and CAS, and is the official journal of The International Neuropsychiatric Association (INA). The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/neuropsychiatric-disease-and-treatment-journal

